Author's response to reviews

Title: The prevalence of underweight, overweight, obesity and associated risk factors among school-going adolescents in seven African countries

Authors:

Taru Manyanga (ummanyan@myumanitoba.ca)
Hesham El-Sayed (heshamfatheyelsayed@gmail.com)
David Teye Doku (daviddoku@ymail.com)
Jason R Randall (jason.randall@ualberta.ca)

Version: 2 Date: 15 September 2013

Author's response to reviews:

I have added a section for the author contributions. I have also added another sentence stating that this study uses publicly available and did not require ethical review from the U of M review board.